VC/Week 39 starting 25 sept 06
SM18 Operation Meeting (Tuesdays) 26 sept 06

Bench status  on Monday  25-09-2006 at 9/25/2006 9:29:03 AM 
	10
	A1
	MBBR2380
	PT 13.2 Resist. Meas.
	298.31

	4
	A2
	MBBR1370
	PT1 IAP@1.9K, Part III
	1.90

	3
	B1
	SSS359
	PT 1 IAP @ cold
	1.90

	11
	B2
	SSS139
	PT 13.2 Resist. Meas.
	298.04

	9
	C1
	SSS144
	ICS 4 Final connection
	296.43

	5
	C2
	SSS236
	Prep 5 Cool Down
	87.77

	2
	D1
	SSS661
	PT 6.6 Training 6
	4.60 HV

	12
	D2
	SSS673
	PT 13.2 Resist. Meas.
	296.79

	6
	E1
	MBAR1376
	Prep 5 Cool Down
	120.23

	7
	E2
	MBBR2375
	ICS 4 Final connection
	297.32

	1
	F1
	MBBL1379
	PT 12 Warm Up
	226.54

	8
	F2
	MBBR2385
	ICS 3 Traveller after warm test
	296.88


 
 
  
CONFIG :  D 6 - AS 5 -  IRSSS 1 ;   
Supervisor:  Gijs -Gilbert  ;Eq support : 16-3602 
15 Magnet Tests completed ( 12 Magnet Stripped ) 
8 Dipoles tests  ( 1334 HV fail)


6Arc-SSS  tests (308 (HV fail) )
 0  500-series [ 526 (192 hr) with thermal cycle]
 1  600-series [608  (245 hrs)] 
General Points:
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· Situation SSSS  :D1 & D2 issues; 
· MM problems on B2; 
· other MM issues/problems this week Sun night …
· A.O.B..
	Magnets under Test from 17-09-2006 to 23-09-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAR1334 
	[image: image1.png]



	


	14-09-2006 
	30 h.
	27 h.
	35 h.
	19 h.
	26 h.
	20-09-2006 
	137 h.
	Stand By
	
	

	2
	TBA2
	MBAR1375 
	[image: image3.png]



	


	17-09-2006 
	28 h.
	29 h.
	14 h.
	26 h.
	10 h.
	22-09-2006 
	107 h.
	Stripping
	
	

	3
	TBA2
	MBBR2377 
	[image: image5.png]



	


	12-09-2006 
	25 h.
	27 h.
	17 h.
	17 h.
	18 h.
	17-09-2006 
	104 h.
	Stripping
	
	

	4
	TBB1
	SSS134 
	[image: image7.png]



	


	15-09-2006 
	19 h.
	14 h.
	18 h.
	7 h.
	7 h.
	18-09-2006 
	65 h.
	Stripping
	
	

	5
	TBB1
	SSS308 
	[image: image9.png]



	


	18-09-2006 
	26 h.
	12 h.
	35 h.
	20 h.
	8 h.
	23-09-2006 
	101 h.
	Stand By
	
	

	6
	TBC1
	SSS338 
	[image: image11.png]



	


	16-09-2006 
	21 h.
	14 h.
	22 h.
	8 h.
	26 h.
	20-09-2006 
	91 h.
	Stripping
	
	

	7
	TBC2
	SSS137 
	[image: image13.png]



	


	19-09-2006 
	37 h.
	20 h.
	15 h.
	8 h.
	12 h.
	23-09-2006 
	92 h.
	Stripping
	
	

	8
	TBC2
	SSS357 
	[image: image15.png]



	


	15-09-2006 
	35 h.
	13 h.
	15 h.
	7 h.
	15 h.
	19-09-2006 
	85 h.
	Stripping
	
	

	9
	TBD1
	SSS302 
	[image: image17.png]



	


	13-09-2006 
	26 h.
	14 h.
	18 h.
	7 h.
	55 h.
	18-09-2006 
	120 h.
	Stripping
	
	

	10
	TBD2
	SSS608 
	[image: image19.png]



	


	08-09-2006 
	108 h.
	20 h.
	39 h.
	16 h.
	62 h.
	18-09-2006 
	245 h.
	Stand By
	
	

	11
	TBE1
	MBBL2346 
	[image: image21.png]



	


	13-09-2006 
	28 h.
	24 h.
	19 h.
	14 h.
	7 h.
	18-09-2006 
	92 h.
	Stripping
	
	

	12
	TBE1
	MBBR2378 
	[image: image23.png]



	


	18-09-2006 
	26 h.
	21 h.
	25 h.
	16 h.
	31 h.
	23-09-2006 
	119 h.
	Stripping
	
	

	13
	TBE2
	MBAL2352 
	[image: image25.png]



	


	14-09-2006 
	25 h.
	26 h.
	20 h.
	15 h.
	24 h.
	19-09-2006 
	110 h.
	Stripping
	
	

	14
	TBF1
	MBAL2280 
	[image: image27.png]



	


	17-09-2006 
	26 h.
	25 h.
	30 h.
	16 h.
	13 h.
	22-09-2006 
	110 h.
	Stripping
	
	

	15
	TBF1
	MBBR1331 
	[image: image29.png]



	


	11-09-2006 
	48 h.
	34 h.
	19 h.
	17 h.
	20 h.
	17-09-2006 
	138 h.
	Stripping
	
	

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR1334
	[image: image31.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR1375
	12360.5 A.
(D1_L)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR2377
	11860.7 A.
(D1_U)
	12691.3 A.
(D1-U)
	12700.7 A.
(D2-U)
	[image: image33.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS134
	11760 A.
(QI_12)
	12070 A.
(QI - 12)
	12161 A.
(QI_34)
	[image: image34.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS308
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS338
	11788.3 A.
(QI_12)
	12798.7 A.
(QE_34)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS137
	12226.3 A.
(QE_12)
	12365.6 A.
(QI_12)
	12562 A.
(QE_12)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS357
	11432 A.
(QI-12)
	11898 A.
(QE--34)
	12307 A.
(QI-12)
	12524.9 A.
(QE-34)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS302
	11778.3 A.
(QI_12)
	12637.9 A.
(QE_12)
	12691.5 A.
(QE_12)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS608
	3488 A.
(E-MQy12)
	3657 A.
(I-MQY12)
	3741.4 A.
(I_MQY_1)
	3786.2 A.
(E_MQY_1)
	3840.8 A.
(I_MQY_2)
	3813.6 A.
(E_MQY_2)
	3858.5 A.
(E_MQY_1)
	3900 A.
[image: image40.jpg]



	-
	-
	-
	-
	-

	11
	MBBL2346
	11242.6 A.
(D2_U)
	11977.4 A.
(D1_L)
	12454.8 A.
(D1_U)
	12638.6 A.
(D2_L)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR2378
	10628.6 A.
(D2_U)
	11366.3 A.
(D1_U)
	11270.2 A.
(D2_U)
	12464.9 A.
(D1_U)
	11621.2 A.
(D1_U)
	12727.8 A.
(D2_U)
	12794.5 A.
(D1-L)
	12859 A.
[image: image42.jpg]



	-
	-
	-
	-
	-

	13
	MBAL2352
	10137.8 A.
(D1_U)
	12027.3 A.
(D2_L)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAL2280
	11082.1 A.
(D1-U)
	11669.4 A.
(D2-U)
	12612.1 A.
(D2_L)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR1331
	11979.8 A.
(D1_L)
	12458.5 A.
(D1_L)
	12670 A.
(D1_U)
	[image: image45.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


[image: image46.png]Magnets from Sunday, September 17, 2006 to Saturday, September 23, 2006
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HV Failures on 1334 & 308;    608 usual issues why connection/ disconnection  so long
 
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_19_M
	TBA1
	MB1334: Temperature too low  for Anti Cryostat. 
	MTM

/ACR
	
	Cryo team informed.

	Tuesday_19_A
	TBF1
	MB2280: AQA does not update MIITS manual. 
MB2280 : During PT5.2 file is not getting transferred So unable to register MIITs values
	MTM
	
	Giorgio asked OP team to proceed and suggested to start the next test after turning LF and HF off.
Giorgio contacted 
Two magnet folders were copied to vol12 via sm18op file manager and not from the one that Giorgio left opened. Then the owner of the directory is sm18op, therefore the automatic copy (incaa based) could not work.

If it is needed to copy manually to vol12, do it from file manager as INCAA. 

	Tuesday_19_N
	TBB2
	SSS139: Due to cryo problem (Cryo1_x2 is not ok), SSW for aperture 2 could not be launched.
The roll angle test was launched. SPA in converter and simultaneously the wire broke. new wire installed
	ACR

MTM
	
	Cryo team is informed. .
Solved after a while.



	Tuesday_19_N

Wednesday_20_M
	TBB2
	SSS139: TCLaux fault.

SSS139: SSW on Aperture 2 Field strength at nominal was launched at about 8:30 after taking clearance form cryo but quench occur during run.
It was also launched at 12:00 but current ramp down due to Cryo instability 
	ACR
	
	Cryo team asked OP team to wait for 30 min. 
Cryo team was informed.

Stéphane suggested to perform SSW Field strength at 5kA. 
Stéphane advised to stop the SSW @cold and continue with the MM due to shortage of time.


	Tuesday_19_N
	TBF1
	MB2280: PT6.1 couldn't be launched on F1 due to error- 222 and bad state 33 of Power Converter. The power converter was not getting ON. 
	MTM
	
	Giorgio will be contacted in the morning.

	Tuesday_19_N
	TBB1
	SSS308: HV failed on Quadrupole Vs Ground step. Others are OK. Report is generated. 
	MTM
	
	Giorgio will be contacted.

Faults confirmed. IFS box removed and investigation by Giorgio and Richard.

	Tuesday_19_M
	TBB2
	SSS139: Temperature too low for Anti Cryostat. 
	MTM


	
	AC cable disconnected for shaft introduction.

	Wednesday_20_A
	TBB2
	SSS139- MM1 error on current reading on Ap1 and Ap2 , error on Temperature reading .
SSS139: Concerning temperature error, wrong bench was selected in magnet settings in MMP. This was first time MM on this bench.
	MTM
	
	 Guy called. Current reading cable was not connected due to previous field advance measurement. Guy connected it, then OK.
Guy corrected the magnet settings, then temperature value correct.

	Wednesday_20_N
	TBB2
	SSS139: Errors in TEMA settings for MM3.
	MTM
	
	Guy corrected the errors and launched the load-line.

	Wednesday_20_N
	TBB2
	SSS139 Quench in (+)Vclamp_ext during MM3 Full load line at 11850A. 
	ACR
	
	Cryo contacted. Quench occurred in Vclamp twice.
Cryo investigation: Condenser problem. It will be solved for the next shift.

	Wednesday_20_N
	TBC1
	SSS244: Error-2 while doing IAP with HF heater discharge. MXI crate reset and restart. The crates were switched off and put on and IAP was done again. The problem repeated. Therefore the crates were put off and restarted again. This time the error was reported after the file was generated. 
	AB/CO
	IAP
	

	Wednesday_20_N
	TBC2
	SSS137: OK to cool-down came after 24 hrs of ICS3 step as cryo leak problem. 

But after 1 hrs of ok to cool down sign, cryo inform that they can not cool down as anticryostat problem.
	ACR
	
	Why? Is there anti-cryostats in SSS137?

	Thursday_21_A
	TBD1
	SSS661: Error in IAP at warm, HF discharge of QH2,QH4, tminV>1000
	MTM
	IAP
	Guy informed. He advised to proceed.

	Thursday_21_A
	TBF2
	MB1342: Leak test not conform. Connection checked at warm by ACR
	ACR
	
	Leak in a flexible inside the CFB. Repaired but magnet to be cooled down again to validate the cryogenic integrity. 

	Thursday_21_A
	TBD1
	SSS661: Temperature of anti-cryostats too low. Alarm displayed
	/MTM
	
	AC cables not connected. 

	Thursday_21_A
	TBB2
	SSS139: Corrector magnetic measurements. The 600A power supply triggered at -300 A.
	AB/PO
	
	AB/PO called. Problem in tuning the power supply parameters?

	Thursday_21_N
	TBB2
	SSS139: MM2 + ramp tests: Error in MM position at high current and low bucking ratio for 1 sector
	MTM
	MMP
	Stéphane informed. Results checked. OK to proceed.

	Friday_22_A
	TBA1
	MB2380: Water inlet point on CFB (blue mark steel pipe) leaks. It sprays when attempt to connect the feed pipes. 
	MTM
	
	The pipe is properly connected after good efforts but repair may be required. Giorgio triggered an action with the mechanic team

	Friday_22_A
	TBA1
	MB2380: Min Energy Quench at 3kA. Error displayed: "Generic error message window" and no LF data available while analysis.
After resetting the crates , no success.
	AB/CO
	lf
	Guy was informed. He asked to reset the crates

	Friday_22_N
	TBA1
	MB2380: PT6.1, message "ERROR, LF cannot be armed" appeared on TEMA display. It couldn't be solved even after resetting the LF & HF crates near test bench..
	AB/CO
	Hf,lf
	Guy was informed. He came and changed the fan unit of HF crate, reset the crates, as well as the sunmta31

	Friday_22_N
	TBB2
	SSS139: HV test for Orbit dipole corrector. Quadrupole vs orbit dipole and quench heater vs ground failed as there was condensation on top of CFB. 
	AB/CO
	Hf,lf
	OP team cleaned water from voltage taps and blower was applied and the repeat steps were ok.

	Sunday_24_N
	TBB1
	SSS359: MXI error during PT1. DAQ (hf, lf) cannot be armed.
	AB/CO
	Hf,lf
	Federico informed in the morning, he killed the process which was hung.

	Monday_25_M
	TBB1
	SSS359: TcL-aux fault. Corrector powering cannot be launched..
	ACR
	
	

































































